Author's response to reviews

Title: Emerging organisational models of primary healthcare and unmet needs for care: insights from a population-based survey in Quebec province.

Authors:

Jean-Frédéric Levesque (Jean-Frederic.Levesque@inspq.qc.ca)
Raynald Pineault (rpineault@santepub-mtl.qc.ca)
Marjolaine Hamel (mhamel@santepub-mtl.qc.ca)
Danièle Roberge (daniele.roberge@usherbrooke.ca)
Costas Kapetanakis (ckapetan@santepub-mtl.qc.ca)
Brigitte Simard (bsimard@santepub-mtl.qc.ca)
Alexandre Prud'homme (aprudhom@santepub-mtl.qc.ca)

Version: 2 Date: 26 October 2011

Author's response to reviews:

Adding ethical approval of our study in the Methods Section of the manuscript.